Author’s response to reviews

Title: Alarming prevalence and clustering of modifiable noncommunicable disease risk factors among adults in Bhutan: a nationwide cross-sectional community survey

Authors:

Dorji Pelzom (dorji.pelzom@gmail.com)
Petros Isaakidis (Petros.Isaakidis@joburg.msf.org)
Myo Oo (dr.myominnoo@gmail.com)
Mongal Gurung (msgurung@health.gov.bt)
Pemba Yangchen (pembayangchen7@gmail.com)

Version: 2 Date: 06 Oct 2017

Author’s response to reviews:

Dear Natalie Pafitis,

Thank you for following your valuable comments,

Editor comments:

(1) Please provide a 'Declarations' section heading after your list of abbreviations.

(2) Please provide a full Declarations section as detailed in our guidelines here: https://bmcpublichealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

This section should include the following sub-sections:

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding

- Authors' Contributions

- Acknowledgements

- Authors' Information

Author’s Response: Thank you for pointing this out and I would like to apologize for having overlooked this in my last revision. I have addressed this and added a ‘Declarations’ section with all the sub-sections at page 17-19 line no.365-409.

Thank you again for your feedback. I look forward to hearing from you soon.

Warm wishes

Dorji Pelzom